Author’s response to reviews

Title: Video feedback and e-Learning enhances laboratory skills and engagement in medical laboratory science students

Authors:

Rebecca Donkin (rdonkin@usc.edu.au)

Elizabeth Askew (easkew@usc.edu.au)

Hollie Stevenson (hsteven1@usc.edu.au)

Version: 2 Date: 03 Aug 2019

Author’s response to reviews:

Thank you for the specific reviewer and editor comments. Please see below for the responses:

1. An extra line has been included in the Discussion to highlight the richer experience and increase in content that the video group voluntarily engaged with.

2. In the "ethical approval and consent to participate" section of the declaration, consent obtained was written and has been included.

3. Figure 1 contains an image of the author RD and Figure 2 contains images of author HS. Both authors have provided written consent for their image to be published in the manuscript.

4. Title has been changed to 'Video feedback and e-Learning enhances laboratory skills and engagement in medical laboratory science students'